<mark>_</mark>



JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2024JG008391

Special Collection:

Quantifying Nature-based Climate Solutions

Key Points:

- Forest structure is primarily shaped by management, but interactions with regional climate change produce divergent structures over time
- Management can increase forest stability and minimize the release of stored carbon by reducing mortality in the face of climate change
- Management-induced changes in resilience are regionally dependent

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

B. A. Murphy, murphyba@ornl.gov

Citation:

Murphy, B. A., Rollinson, C. R., Dietze, M. C., Staudhammer, C. L., VonHedemann, N. R., Schultz, C. A., et al. (2025). Insights into nature-based climate solutions: Managing forests for climate resilience and carbon stability. *Journal of Geophysical Research: Biogeosciences*, *130*, e2024JG008391. https://doi.org/10. 1029/2024JG008391

Received 4 AUG 2024 Accepted 14 FEB 2025

Author Contributions:

Conceptualization: Bailey A. Murphy, Christine R. Rollinson, Michael C. Dietze, Christina L. Staudhammer, Courtney A. Schultz, William J. Kleindl, Ankur R. Desai Data curation: Bailey A. Murphy, Christine R. Rollinson, Nicolena R. VonHedemann Formal analysis: Bailey A. Murphy Funding acquisition: Michael C. Dietze, Christina L. Staudhammer, Courtney

© 2025 Oak Ridge National Laboratory and The Author(s). This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Insights Into Nature-Based Climate Solutions: Managing Forests for Climate Resilience and Carbon Stability

Bailey A. Murphy^{1,2}, Christine R. Rollinson³, Michael C. Dietze⁴, Christina L. Staudhammer⁵, Nicolena R. VonHedemann⁶, Courtney A. Schultz⁶, William J. Kleindl⁷, and Ankur R. Desai¹

¹Department of Atmospheric and Oceanic Sciences, University of Wisconsin-Madison, Madison, WI, USA, ²Now at Oak Ridge National Laboratory, Environmental Sciences Division and Climate Change Science Institute, Oak Ridge, TN, USA, ³Center for Tree Science, The Morton Arboretum, Lisle, IL, USA, ⁴Department of Earth and Environment, Boston University, Boston, MA, USA, ⁵Department of Biological Sciences, University of Alabama, Tuscaloosa, AL, USA, ⁶Department of Forest and Rangeland Stewardship, Colorado State University, Fort Collins, CO, USA, ⁷Department of Land Resources and Environmental Sciences, Montana State University, Bozeman, MT, USA

Abstract Successful implementation of forest management as a nature-based climate solution is dependent on the durability of management-induced changes in forest carbon storage and sequestration. As forests face unprecedented stability risks in the face of ongoing climate change, much remains unknown regarding how management will impact forest stability, or how interactions with climate might shift the response of forests to management across spatiotemporal scales. Here, we used a process-based model to simulate multidecadal projections of forest dynamics in response to changes in management and climate. Simulations were conducted across gradients in forest type, edaphic factors, and management intensity under two alternate radiative forcing scenarios (RCP4.5 and RCP8.5). This allowed for the quantification of forest stability shifts in response to climate change, and the role of management in modulating that response, where ecosystem stability is characterized as the resilience and temporal stability of net primary production, aboveground biomass, and soil carbon. Our results indicate that forest structure is primarily shaped by management, but the same management strategy often produced divergent structures over time, due to interactions with regional climate change. We found that management can be used to increase stability and minimize the release of stored carbon by reducing mortality, but also highlight the regional dependency of management-induced changes in resilience to climate change.

Plain Language Summary Successfully using forest management to help mitigate climate change depends on how well management can keep forests storing and absorbing carbon dioxide over time. As climate change continues, forests are facing new and serious risks, and we do not fully understand how management efforts will affect their stability. It is also unclear how climate changes might alter the way forests respond to management. In this study, we used a computer model to predict how forests might change over several decades due to different management practices and climate conditions. We looked at various forest types and levels of management intensity under two possible climate scenarios (moderate and high levels of greenhouse gas emissions). We measured stability in terms of how resilient forests are and how stable their growth, biomass, and soil carbon levels remain over time. The results showed that management practices have a significant impact on forest structure, but that the same management approach can lead to different outcomes depending on regional climate changes. We also found that management can improve forest stability and reduce carbon loss by lowering tree mortality rates. However, the effectiveness of these strategies heavily depends on the specific regional climate conditions.

1. Introduction

Emission reductions alone are insufficient to avoid the most catastrophic effects of climate change (Canadell & Schulze, 2014; National Academies of Sciences, Engineering, and Medicine, 2019; Rockström et al., 2021). Therefore, the active removal of atmospheric CO_2 must also be a core component of any feasible climate mitigation strategy (IPCC, 2022; Smith et al., 2016). Many proposed approaches to climate change mitigation require significant technological advances or are still largely experimental (Fuss et al., 2014). Nature-based climate solutions (NbCS), which manipulate natural systems through management and design to increase carbon sequestration and decrease greenhouse gas emissions (Novick et al., 2022), have emerged as prospective solutions



A. Schultz, William J. Kleindl, Ankur R. Desai

Investigation: Bailey A. Murphy Methodology: Bailey A. Murphy Project administration: Bailey A. Murphy, Ankur R. Desai Resources: Christine R. Rollinson. Michael C. Dietze, Christina L. Staudhammer, Ankur R. Desai Supervision: Ankur R. Desai Validation: Bailey A. Murphy Visualization: Bailey A. Murphy Writing - original draft: Bailey A. Murphy Writing - review & editing: Bailey A. Murphy, Christine R. Rollinson, Michael C. Dietze, Christina L. Staudhammer, Nicolena

R. VonHedemann, Courtney A. Schultz, Ankur R. Desai that are less reliant on emergent technology. NbCS have the potential to sequester up to 1.2 Pg CO2 e year⁻¹ in the United States (U.S.), which is approximately 21% of net annual emissions (Fargione et al., 2018), while also supporting a range of environmental and social cobenefits and potentially increasing resilience to future climate change impacts (Becknell et al., 2015; Fargione et al., 2018; Novick et al., 2022).

Successful implementation of forest management as a NbCS is dependent on the durability of managementinduced changes in forest carbon storage and sequestration (Anderegg, Chegwidden, et al., 2022; Canadell & Raupach, 2008; Novick et al., 2022). However, recent evidence indicates that forest carbon sinks are becoming more vulnerable in response to the increased variability and magnitude of extremes in climate drivers (Forzieri et al., 2022; McDowell et al., 2020; Wu et al., 2023). Rapidly changing environmental conditions have the potential to decrease forest stability (Bauman et al., 2022; Reich et al., 2022); the resistance to changes in forest function or composition in response to disturbance, or the rate, variability, and extent of the return to predisturbance function or composition (i.e., resilience) (Hillebrand et al., 2018; Holling, 1973; Mathes et al., 2021; Tilman et al., 2006). As forests become less stable, the likelihood of large-scale mortality in response to climatedriven disturbances increases (Bauman et al., 2022; Forzieri et al., 2022; Reich et al., 2022), meaning that a significant portion of stored carbon could be released, potentially nullifying mitigation efforts (Anderegg, Chegwidden, et al., 2022; Reichstein et al., 2013). For example, Running (2008) estimated that Hurricane Katrina killed approximately 320 million trees, turning the Southeastern (SE) U.S. into a net carbon source that offset the entire annual U.S. terrestrial carbon sink in 2005.

Forest structure plays a critical role in determining the response of forests to climate stressors (Amiro et al., 2010; Anderegg, Chegwidden, et al., 2022). Variations in forest structure such as tree density, species composition, and tree size and age distributions influence forest stability by shaping resource availability and carbon dynamics (Anderson-Teixeira et al., 2013; Gough et al., 2019; Matheny et al., 2014), and targeted management has the potential to promote forest stability through structural manipulations that mitigate disturbance risk. For example, fuels reduction management (including thinning and prescribed fire) is frequently used to reduce the risk of severe fire and associated mortality (Hurteau et al., 2014), and a study by Brice et al. (2020) showed that moderate reductions in stand basal area reduced turnover and convergence times associated with forest transition, accelerating climate-driven biome shifts. However, fewer studies have examined the ability of management to moderate forest responses to climate change as a press disturbance across multidecadal timescales, while also accounting for climate feedback, despite the necessity to ensure that forest management NbCS are robust and scalable. Additionally, the evaluation of forest management as a NbCS has primarily focused on extended rotations as a management strategy (Fargione et al., 2018; Kaarakka et al., 2021), but other commonly applied silvicultural practices such as thinning (either of the understory or large tree removal) and selective harvest could enhance forest carbon sequestration and stability. Addressing this knowledge gap requires models that include climate sensitivity in forest dynamics projections, which current NbCS assessment tools, such as the Forest Vegetation Simulator (Dixon, 2002), do not support. Recent proposals on NbCS scientific priorities (Anderegg, Trugman, et al., 2022) underscored the need to integrate demographic models incorporating plant physiology, ecological dynamics, and climate sensitivity in the assessment process to ensure that NbCS are rooted in rigorous scientific understanding and emphasize that we must account for how climate feedback could alter carbon dynamics and disturbance responses (Novick et al., 2024).

Here, we used a process-based vegetation demography model to simulate multidecadal projections of forest dynamics in response to changes in management and climate in two regions: the Great Lakes (GL) and the SE U.S. Simulations span from 2006 to 2100 and are conducted under two Representative Concentration Pathways (RCP4.5 and RCP8.5) and three management strategies (even-age, high-intensity uneven-age, and low-intensity uneven-age) capturing a range of intensities in applied management practices. This enables the evaluation of forest stability shifts due to climate change, the role of management in modulating these shifts, and regional environmental variation. This study focused on the stability of existing carbon stored in aboveground biomass (AGB) and soils (referred to as compositional stability), as well as on the ability of the forest to sequester carbon (referred to as functional stability), represented by net primary productivity (NPP). Forest stability is characterized here as both resilience (speed of functional recovery following perturbation; Holling, 1973) and temporal stability (persistence of ecosystem function over time; Lehman & Tilman, 2000; Tilman et al., 2006). Several studies have adopted a multidimensional framework to evaluate stability in response to discrete disturbance events, such as stem girdling through the Forest Accelerated Succession Experiment (FASET; Gough et al., 2013; Mathes et al., 2021), or simulated mortality events across a range of severities (Dorheim et al., 2022). However,

2 of 19

this study is among the first to evaluate the multidimensional stability response of forest carbon cycling to sustained management over multidecadal timescales and within the context of alternate climate change scenarios at the regional scale. Therefore, this work provides novel insights into how management impacts the structure and stability of forests across multidecadal timescales and how future climate change could potentially affect stability trajectories.

Specifically, we address the following questions: (a) How does management shape forest structure across multidecadal timescales, and do impacts differ depending on the emissions scenario? (b) How is the stability of forest carbon storage and sequestration affected by changes in forest structure, as shaped by management and climate, and are impacts regionally dependent? We hypothesize that forest structure is primarily shaped by management, and that the impact of management on structure will depend on management intensity. However, we also expect that structural responses to management will diverge regionally over time because of differences in regional climate change, particularly the significantly increased precipitation anticipated in the SE. We hypothesize that management is the most important factor shaping functional stability and that climate is the primary driver of compositional stability; however, regional interactions between management and climate change will alter stability outcomes over time and that more intensive management practices will decrease forest stability. Finally, we expect to see a positive correlation between forest stability and structural complexity, indicating that managing for enhanced structural complexity could promote forest stability in the face of ongoing climate change.

2. Methods

2.1. Site Description

National Ecological Observatory Network (NEON) core terrestrial sites in the GL and SE U.S. were selected for this study. National Ecological Observatory Network core terrestrial site locations were determined based on their representativeness of the U.S. ecological landscape along multiple axes of climate, edaphic, topographic, vegetation, wildlife, and management practices, using a statistical clustering algorithm (Schimel et al., 2007). The two regions chosen here represent a range of forest types and environmental conditions, and the use of NEON sites ensured ample data availability for model initialization. The University of Notre Dame Environmental Research Center (UNDE, 46.23°N, 89.54°W), located in Michigan's Upper Peninsula, represents the GL region. UNDE, a northern mesic forest site, features primarily second-growth forest with a history of clear-cutting in the early twentieth century (Mahon, 2003). UNDE experiences an average annual temperature of 4.5°C, 800 mm of average yearly precipitation, and notable winter snowfall. The soil, mainly Spodosols with some Histosols and Inceptisols, has sandy textures and originated in glaciofluvial deposits (Parsley, 2016). The area is characterized by forests, lakes, ponds, and bogs, with dominant tree species including red and sugar maple (*Acer rubrum* and *A. saccharum*, respectively), aspen (*Populus tremuloides* and *P. grandidentata*), and paper birch (*Betula papyrifera*) (Krauss, 2018).

The SE U.S. is represented by the Ordway-Swisher Biological Station (OSBS, 29.69°N, 81.99°W) and the Talladega National Forest (TALL, 32.95°N, 87.39°W) sites. Ordway-Swisher Biological Station, a longleaf pine site in North central Florida, experiences a humid subtropical climate with an average annual temperature of 21°C and an average precipitation rate of 1,300 mm per year. The soils are sandy, and impacted by silviculture (Prink & Figueroa, 2019). Dominant tree species include longleaf pine (*Pinus palustris*) and turkey oak (*Quercus laevis*) (Krauss, 2018), and the site is managed for recreation, habitat, and biodiversity, utilizing prescribed fires (Livingston, 2014). TALL, located in West Central Alabama, is predominantly longleaf and loblolly pine (*P. palustris*) and *P. taeda*) with an understory of mixed oak and hardwoods (Hatcher, 2017). The site has a humid subtropical climate with an average annual temperature of 17°C and an average precipitation rate of 1,400 mm per year. Soils, developed from marine sediments, include sandy, loamy, and clayey types (Hatcher, 2017). TALL is managed for recreation, active logging, preservation, and habitat, involving regular midstory harvests and prescribed fires (Pasquill, 2006).

2.2. Ecological Modeling

2.2.1. Model Description

This study utilizes the Ecosystem Demography model version 2.1 (ED2; Moorcroft et al., 2001; Hurtt et al., 2002; Albani et al., 2006; Medvigy et al., 2009; Longo et al., 2019). ED2 is a vegetation dynamics model that simulates



population dynamics, demography, and community composition using a hierarchical structure. Rather than simulating individual trees, ED2 simulates cohorts–groups of trees with similar size and age characteristics–making it less computationally expensive than traditional forest gap models (Fisher & Koven, 2020) while maintaining its ability to represent complex ecological interactions. Cohorts in ED2 compete for available resources and are distributed across patches, each representing a spatially implicit microenvironment with a shared disturbance history (Longo et al., 2019). Patches are dynamic, meaning their fractional area changes in response to aging and disturbance. ED2's cohort-based approach facilitates harvest based on tree size and age, and its patch structure enables the evolution of forest structure, community composition, and microenvironment in response to disturbance. This makes ED2 uniquely well suited for exploring the impact of management and climate change on forest dynamics.

2.2.2. Model Simulations

Simulations used downscaled air temperature, precipitation, incoming shortwave and longwave radiation, specific humidity, air pressure, and wind speed from an ensemble of 10 Coupled Model Intercomparison Project Phase 5 (CMIP5; Taylor et al., 2012) general circulation models (GCMs) (Table S1 in Supporting Information S1), for the period 2006–2100. A two-stage generalized additive modeling (GAM) approach (LeBauer et al., 2013; Simkins, 2017) downscaled meteorological data from daily to the subdaily resolution required by ED2 (see supplement S1 Section 1 for details). General circulation model output from two radiative forcing scenarios, RCP4.5 and RCP8.5, was used to explore forest responses to different climate futures. RCP4.5 represents a moderate emissions pathway involving significant mitigation, while RCP8.5 represents a business-asusual scenario with unchecked emissions (van Vuuren et al., 2011). These scenarios cover a range of emissions trajectories, facilitating comparisons to other studies. Simulations were initialized with site-specific forest composition and structure data under a stable CO_2 concentration (380 ppm), corresponding to the atmospheric CO_2 concentration in 2006, when model simulations began. This avoided compounding effects of enhanced CO_2 on plant productivity (Bacastow & Keeling, 1973; Ciais et al., 2013; Walker et al., 2020), as potential CO₂ fertilization effects were not a focus of this study. ED2's sensitivity to CO₂ increases can result in unrealistic plant productivity responses and dominate prediction uncertainty over long timescales (De Kauwe et al., 2013, 2014; Rollinson et al., 2017; Walker et al., 2020; Zaehle et al., 2014).

Soil and vegetation data to initialize ED2 came from NEON plot and site-level observations. Soil characteristics, including texture, carbon and nitrogen content, bulk density, and microbial biomass, were obtained from megapit data products (DP1.00096.001, DP1.10104.001, and DP1.10098.001). Additional information on soil carbon (SOC) pool initialization is provided in the supplement. In situ measurements of tree density, height, diameter at breast height (DBH), and tree species at the plot scale were used to initialize vegetation. Tree species were converted to seven plant functional types (PFTs) recognizable by ED2, where PFTs are groups of species with similar physical, phylogenetic, and phenological characteristics (Wullschleger et al., 2014): a temperate C3 grass; southern pine; northern pine; late successional conifer; and early, mid, and late successional hardwood tree PFTs. All tree PFTs used default ED2 trait values; site-specific parameterization was not performed.

Management in simulations was represented as one of three types: even-aged, low-intensity uneven-aged, and high-intensity uneven-aged, reflecting a range of practices. Even-age management involves full or selective periodic timber harvest with minimal management between harvests, common among nonindustrial private forests (Carey, 2006). Due to model constraints, some age variability persists in even-age management because harvest is prescribed as a fraction of the total area each year, so rather than a "true" clear-cut, it is more akin to patch clear-cuts. Uneven-age management involves more frequent harvests targeting trees above a minimum marketable diameter. High-intensity uneven-aged forestry involves harvest practices that remove large fractional areas of trees with shorter return intervals than the low-intensity uneven-aged management scenario and incorporate understory thinning. Low-intensity uneven-aged management occurs on longer rotations and involves some understory thinning in the GL and regeneration support in the SE. A management-free control scenario provided a baseline for comparison. Management in ED2 is prescribed by altering the spatial extent of harvest and the size groups (tree diameter) and PFTs targeted for removal. Harvest rotations are implicitly represented by specifying a fractional area of trees to remove each year, calculated as 1/harvest rotation. Regional management parameters were derived from interviews with forestry experts in the regions of interest (von Hedemann and Schultz, 2021). For example, from these discussions, high-intensity uneven-age forestry in the GL is represented in ED2 with a 57-year rotation, which translates to a fractional harvest area of 1.8% per year, where trees with a

DBH above 26.67 cm have an 80% probability of harvest, and trees below have a 20% probability of harvest. To allow vegetation and climate forcing to equilibrate, simulation of management did not begin until 2020. Additional information regarding the derivation of management parameters is provided in the supplement, with detailed regional parameter values in Table S2 in Supporting Information S1.

Although ED2 can simulate natural disturbances, they were not explicitly represented to isolate the impact of management on functional and compositional responses to climate change. Simulations started from existing forest conditions rather than bare ground, so a prolonged spin-up period to build up ecosystem carbon pools was not necessary. Carbon fluxes simulated under the control scenario were compared against NEON-supported eddy covariance flux tower observations from UNDE, TALL, and OSBS (see supplement S1 Section 5).

2.3. Management and Climate Impacts on Forest Structure

Structural variables' output by ED2 were evaluated under each management and emissions scenario at the start and end of the century to determine average changes over time. Forest structure was characterized by nine variables: tree age (years), leaf area index (LAI), tree density (trees m⁻²), the mean and standard deviation of tree DBH (\overline{DBH} and σ_{DBH} , cm), crown area (\overline{CA} and σ_{CA} , m), and tree height (\overline{H} and σ_{H} , m). The standard deviation of tree diameter, crown area, and height represent variability in tree size and indicate canopy complexity, which has demonstrated links to forest productivity (Atkins et al., 2018, 2020; Gough et al., 2019; Hardiman et al., 2013; Murphy et al., 2022). All structural metrics natively exist at the cohort or patch scale in ED2 and were scaled to the site level using area-weighted averaging. Seedlings and saplings smaller than 12.7 cm in diameter and 1.37 m in height were excluded to remove bias from overrepresented seedlings, based on Forest Inventory and Analysis thresholds. One-way Kruskal-Wallis tests determined if forest structure differed significantly between management types within each region, and Dunn's post hoc test was applied to determine the significance for pairwise comparisons of individual structural variables. A Bonferroni adjustment was applied to p-values.

2.4. Forest Stability

This study evaluated the impact of forest management and climate change on forest stability, characterized as resilience and temporal stability. Resilience is the capacity of ecosystems to recover from perturbations (Holling, 1973), often represented as the rate of return to predisturbance function or composition (Hillebrand et al., 2018; Mathes et al., 2021), while temporal stability measures variability in function or composition relative to the mean state (Lehman & Tilman, 2000; Tilman et al., 2006). We assessed these impacts on three key variables: AGB, SOC, and net primary production (NPP). AGB and SOC represent existing carbon stores (compositional variables), and NPP represents ongoing carbon sequestration (functional variables). Additional information on representation of SOC in ED2 is provided in the supplement.

Monthly model output was the subset to the growing season, defined as periods with air temperatures above 5° C and gross primary productivity over 0.03 kg C m⁻² month⁻¹ (Nelson et al., 2018). This definition of growing season was employed to accommodate regional differences in growing season length, and potential shifts in phenological timing that might occur due to climate change. Data were grouped by region, emissions scenario, management type, and GCM to calculate ratios between treatment and control values, and then averaged across GCMs to calculate annual resilience values. Temporal stability was determined over nonoverlapping 5-year intervals from the start of active management in 2020–2100. Data were organized into start (2020–2059) and end of century (2060–2100) bins and averaged to calculate net changes in forest structure, resilience, and temporal stability.

Stability was calculated using control scenarios under RCP4.5 and RCP8.5 and comparing them to simulations under management and climate treatments, where each treatment denotes a combination of the management type (even-age, low-intensity uneven-age, or high-intensity uneven-age forestry) or the management-free control scenario, and emissions scenario (RCP4.5 or RCP8.5), for a total of eight treatments applied in each region. Each treatment consists of 10 replicates; model output from each of the 10 GCMs represents a treatment replicate. Resilience was calculated following Hillebrand et al. (2018). Functional resilience is the regression slope of relative NPP over time (Equation 1), and compositional resilience, or the resilience of carbon stored as AGB and in soils, is the slope of similarity between control- and treatment group-stored carbon magnitude over time (Equation 2).



21698961, 2025, 3, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024JG008391, Wiley Online Library on [03/06/2025]. See the Terms

Equation 1 Mathematical representation of ecosystem functional resilience from Hillebrand et al. (2018). F_{Perturbed} is the perturbed function value, $F_{Control}$ is the control function value, *i* is the intercept of the resilience regression line, t is time, and F_R is functional resilience.

$$ln\left(\frac{F_{Perturbed}}{F_{Control}}\right) = i + F_R * t \tag{1}$$

Equation 2 Mathematical representation of compositional resilience (Hillebrand et al., 2018), where compositional resilience refers to the resilience of stored carbon. C_{Perturbed} is the perturbed compositional value, C_{Control} is the control compositional value, i is the intercept of the resilience regression line, t is time, and C_R is compositional resilience.

$$sim\left(\frac{C_{Perturbed}}{C_{Control}}\right) = i + C_R * t \tag{2}$$

The similarity measure used in Equation 2 is Euclidean distance, the straight-line distance between two points in n-dimensional space, a measure frequently used for continuous numerical data. Similarity values between zero and one are possible, where values closer to one indicate a greater degree of similarity between two numbers. Resilience is a dimensionless quantity and can be positive, negative, or zero. Zero indicates no recovery, positive indicates rapid recovery, and negative indicates lower resilience than the control in response to forcing or perturbation (Hillebrand et al., 2018). Calculating resilience from the log ratio and similarity measures of perturbation to control values standardizes responses for comparison across variables with different units and magnitudes. Log-transformed resilience values were scaled by two orders of magnitude to facilitate interpretation of treatment differences. Scaling does not affect the relationships or significance of results but provides a more interpretable range of values. Temporal stability was calculated as the inverse of the coefficient of variation of the functional or compositional variable over a set timestep (Equation 3; Tilman et al., 2006; Lehman & Tilman, 2000; Tilman, 1999). This measure of temporal stability was used rather than the inverse of the standard deviation of resilience residuals (Hillebrand et al., 2018; Mathes et al., 2021), as it is more applicable when measuring stability over long time periods in the face of multiple fluctuating disturbances (Tilman et al., 2006), as opposed to capturing stability responses to single-pulse disturbance events (Hillebrand et al., 2018). Temporal stability is a dimensionless quantity that is inherently positive; larger values correspond to lower fluctuations around the mean trend in ecosystem function and composition over time. Temporal stabilities were logtransformed prior to analyses to facilitate comparison across variables with different units.

Equation 3 Mathematical representation of temporal stability, from Tilman et al. (2006). S_T is temporal stability, μ is the mean functional value over a set time period, and σ is the standard deviation of the functional value over the same time period. Temporal stability values are log-transformed to ensure normality and to facilitate comparison across variables with different units.

$$S_T = ln\left(\frac{\mu}{\sigma}\right) \tag{3}$$

A GAM, implemented using the R package mgcv version 1.8.42 (Wood, 2011) and fit using the restricted maximum likelihood method, assessed whether ecosystem stability was better predicted by management type or climate change scenario. Region was included as a predictor to control for inherent location-specific differences between the three study sites. This ensures that observed effects of management and climate change scenarios are not confounded by the substantial variability between site locations. Interactive effects between management type, region, and emissions scenario were included as predictors, with year included as a smoothed predictor using thin plate regression splines to account for temporal autocorrelation. Spearman correlation determined the directionality and strength of relationships between stability metrics and forest structure. Statistical significance of stability differences across management types, geographic regions, and emissions scenarios was tested using Tukey's Honestly Significant Difference (HSD) test with an alpha of 0.05. A Bonferroni adjustment was applied to p-values. All analyses were performed in R (R Core Team, 2021).

and-conditions) on Wiley Online Library for rules of use; OA

are governed by the applicable Creative Commons License



3. Results

3.1. Regional Climate, Productivity, and Stored Carbon Trends

Average monthly growing season precipitation and temperature increased from the start to the end of the century in both regions. Precipitation increased substantially in the SE, intensifying by 16%–20%, whereas increases in the GL were negligible (1%–2%). Precipitation increases were more pronounced under RCP4.5, and regional differences in precipitation were larger under the higher emissions scenario. Temperature increases were greatest under the high-emissions scenario (0.4° C– 0.9° C increase under RCP4.5 compared to 0.6° C– 1.9° C increase under RCP8.5), and regional differences became more pronounced over time. These simulated changes in precipitation and air temperature align with regional trends reported in the Fourth National Climate Assessment (USGCRP, 2018).

Productivity was highest in the GL, although regional differences narrowed over time (Figure S3 in Supporting Information S1), and NPP was higher on average under RCP8.5 than RCP4.5. For RCP4.5, productivity was highest under low-intensity even-age management in both regions (Figure S3 in Supporting Information S1). Productivity did not differ significantly by the management type under RCP8.5 in the GL, and NPP was highest under even-age and high-intensity uneven-age management in the SE. In addition to higher productivity, carbon stores were larger in the GL. Soil carbon stores were approximately twice as large in the GL (64.02–65.80 kg C m⁻² under RCP4.5 and 59.95–62.85 kg C m⁻² under RCP8.5), with the highest values observed under low-intensity uneven-age management, while in the SE, SOC averaged 31.24–32.65 kg C m⁻² under RCP4.5 and 32.79–35.34 kg C m⁻² under RCP8.5. The amount of carbon stored as AGB averaged 13.09–17.45 kg C m⁻² under RCP4.5 and 13.55–17.76 13.09–17.45 kg C m⁻² under RCP8.5 in the GL, with the highest values under low-intensity uneven-age management. In the SE, AGB averaged 10.55–12.72 kg C m⁻² under RCP4.5 with the highest values under low-intensity uneven-age management. In the SE, AGB averaged 10.55–12.72 kg C m⁻² under RCP8.5, with the highest values under low-intensity uneven-age management. In the SE, AGB averaged 10.55–12.15 kg C m⁻² under RCP8.5, with the highest values under low-intensity uneven-age management.

3.2. Management and Climate Impacts on Forest Structure

Forest structure differed significantly between management types in the SE, but less so in the GL, where differences between the two intensity levels of uneven-age management scenarios were often not significant. Generally, structural differences were more pronounced between management types than between emissions scenarios, although emissions-related differences in mean tree height and size variability (σ_H , σ_{DBH} , and σ_{CA}) were notable. However, the impact of management strategies varied regionally, often producing divergent structural outcomes over time. Impacting forest structure, tree mortality rates increased by 70.25%–98.21% from the start to the end of the century, with the highest mortality rates observed under the control scenario. All active management strategies reduced tree mortality compared to the control. Even-age management had the lowest mortality rates in the GL, while high-intensity uneven-age forestry had the lowest mortality rates in the SE. In the GL, slightly higher mortality rates were associated with a more severe emissions scenario, but the opposite trend was observed in the SE.

In the absence of management, trees in the SE tended to have larger diameters, higher LAI, and greater size variability (σ_{DBH} and σ_H) than trees in the GL by the end of the century, while trees in the GL had higher average crown size and variability (\overline{CA} and σ_{CA}) and were more densely populated; however, average tree age and height was similar between the two regions (Table S4 in Supporting Information S1). Tree age, size (\overline{DBH} and \overline{H}), and variability in size (σ_{DBH} and σ_H) increased over time in both regions. LAI and variability in crown size (σ_{CA}) increased over time in the SE, but changes were minimal in the GL. Density decreased over time in the GL but exhibited minimal change over time in the SE. By the end of the century, active management resulted in younger trees with reduced size variability (σ_H and σ_{DBH}) than the control in both regions under RCP4.5 (Table S4 in Supporting Information S1). Even-age management decreased tree height and diameter relative to the control in both regions (Figure S1 in Supporting Information S1) but increased average crown size in the SE, and decreased variability in crown size in the GL but did not differ significantly from the control in the SE under RCP4.5. Even-age management resulted in the youngest average stands in the GL (Figure 1). Tree density and LAI did not differ significantly from the control scenario in either region under even-age management (Figure 1, Figure S2 in Supporting Information S1). Low-intensity uneven-age management decreased average crown size and EAI did not differ significantly from the control scenario in either region under even-age management (Figure 1, Figure S2 in Supporting Information S1). Low-intensity uneven-age management decreased average crown size and





Figure 1. Average age, leaf area index (LAI), and standard deviation of tree height (σ_H) for the start and the end of the century. Columns are organized by geographic regions (a, c, e are GL and b, d, f are SE); color hue corresponds to the management type, and color tone corresponds to climate change scenario, where lighter tones represent RCP4.5 and darker tones represent RCP8.5. Within a region, emissions scenario, and century, "*" indicates whether distributions under a given management type differed significantly from the control scenario at an alpha level of p < 0.05.

variability relative to the control in both regions (Figures S1 and S2 in Supporting Information S1) and resulted in lower LAI and tree density in the SE and taller trees in the GL. Low-intensity uneven-age management did not significantly impact \overline{DBH} in either region under RCP4.5, meaning tree \overline{DBH} was comparable to the control (Figure S1 in Supporting Information S1). High-intensity uneven-age management decreased crown size and variability relative to the control in both regions and decreased tree size (\overline{DBH} and \overline{H}) in the SE, but differences were not significant in the GL. LAI did not differ significantly from the control scenario in either region under high-intensity uneven-age management for RCP4.5 (Figure 1).

Structural variables representing tree height and variability in tree size (σ_H , σ_{DBH} , and σ_{CA}) were particularly sensitive to differences in climate by the end of the century (Table S4 in Supporting Information S1). Tree heights were significantly different by emissions scenario for all management types in the GL, and for high-intensity uneven-age forestry in the SE (Figure S1 in Supporting Information S1). Average tree height increased for all management types under RCP8.5 in the GL, and for high-intensity uneven-age forestry in the SE. However, height differences between emissions scenarios under active management were less pronounced than height differences between emissions scenarios for the management-free control in both regions, suggesting management effectively reduced the climate sensitivity of tree height to some degree. Variability in tree size (σ_H , σ_{DBH} , and σ_{CA}) was generally lower under RCP8.5 in both regions. Active management minimized the effect of emissions scenario on height variability relative to the control in both regions (Figure 1). Even-age management in the GL and high-intensity uneven-age management in the SE exhibited the lowest height variability climate sensitivity. Even-age management in the SE exhibited the lowest height variability climate sensitivity. Even-age and high-intensity uneven-age management minimized emissions-driven differences in σ_{DBH} relative to the control in the SE, but low-intensity uneven-age management increased the climate sensitivity of σ_{DBH} . High-intensity uneven-age management dampened the response of σ_{CA} to climate intensification in the GL, but even-age management was the most effective in the SE.





Figure 2. Mean annual aboveground biomass (AGB; panels a and d), net primary productivity (NPP; panels b and e), and soil carbon (SOC; panels c and f) resilience by management type and emissions (RCP) scenario. Individual points are annual averages, and point shapes correspond to emissions scenarios. Black points are overall group means from 2020–2100 (error bars denote $\pm 1 \sigma$). For clarity, NPP resilience values >5 and <-5 are not shown here, but contributed to group means. Letters indicate statistically significant differences in resilience by the management type, evaluated within each region and emissions scenario. Groups sharing the same letter are not significantly different from each other, while groups with different letters are considered statistically significant at an alpha level of *p* < 0.05. Letters A, B, and C correspond to RCP4.5, and letters D, E, and F correspond to RCP8.5.

3.3. Management and Climate Impacts on Forest Stability

3.3.1. Resilience

Functional (NPP) resilience differed significantly between even-age and uneven-age management in the GL but did not differ within uneven-age management types (Figure 2). In the SE, NPP resilience differed significantly between even-age and low-intensity uneven-age management. The impact of specific management strategies on functional resilience was regionally dependent. For example, uneven-age management lowered functional resilience relative to the control in the GL but enhanced it in the SE (Figure 2). Even-age management had the best functional resilience outcomes in the GL, while low-intensity uneven-age forestry had the greatest positive impact in the SE (Table S5 in Supporting Information S1). Functional resilience decreased from the start to the end of the century in both regions and was higher on average in the SE (Table S5 in Supporting Information S1). Functional resilience did not differ significantly between emissions scenarios within management types.

The impact of individual management strategies on compositional resilience was also regionally dependent. In the SE, SOC compositional resilience differed significantly between all management types under RCP4.5, and AGB resilience under low-intensity uneven-age management differed significantly (Figure 2). In the GL, SOC resilience differed between the two uneven-age management strategies under RCP4.5, and differences in AGB resilience were significant for all management types under RCP4.5. However, under RCP8.5, differences in compositional resilience by the management type were not significant in either region (Figure 2). High-intensity uneven-age and even-age management had the greatest positive impact on compositional resilience in the SE, enhancing resilience relative to the control, but the magnitude of enhancement depended on the emissions scenario. In the GL, the resilience of stored carbon was typically lower under the higher emissions scenario, although the region experienced decreased compositional resilience and reductions in total AGB and SOC over time under both emissions scenarios (Figure S3 in Supporting Information S1). Conversely, in the SE, compositional

are governed by the applicable Creative Commons License





Figure 3. Average start and end of the century temporal stability by region (point shape), management type (color), and RCP scenario (line type) for (a) aboveground biomass (AGB), (b) net primary productivity (NPP), and (c) soil carbon (SOC).

resilience was higher on average under RCP8.5, with considerable increases in aboveground carbon stores over time.

3.3.2. Temporal Stability

The temporal stability of productivity was notably lower than that of stored carbon stocks due to the high interannual variability in NPP (Table S6 in Supporting Information S1). Regional differences in temporal stability were more pronounced for stored carbon stocks (AGB and SOC), compared to functional temporal stability (NPP, Figure 3). Compositional temporal stability was generally higher in the GL, although SOC temporal stability differences decreased over time as SOC declined steadily in the GL during the second half of the century. AGB temporal stability increased slightly from the first to the second half of the century in both regions, with differing trends in AGB magnitude. Overall, AGB decreased in the GL and increased in the SE, but both became less variable in time, resulting in increased temporal stability.

Differences in functional temporal stability between management types were insignificant, but significant differences were observed in compositional temporal stability between high-intensity uneven-age and even-age management, as well as between high- and low-intensity uneven-age management in the SE. AGB temporal stability was highest under low-intensity uneven-age and even-age management in the SE, while SOC temporal stability was highest under even-age management in the SE. Temporal stability differed significantly between emissions scenarios in both regions.

3.4. Predictors of Ecosystem Stability

The GAM predicting forest stability as a function of time, management type, emissions scenario, and region (including interactive effects) showed that after accounting for baseline differences across locations, the interaction term between management and the region was the primary driver of functional resilience ($\eta_P^2 = 0.04$) and compositional temporal stability ($\eta_P^2 = 0.02$) (Table 1), while the interaction term between emissions scenario and the region was the primary driver of compositional resilience ($\eta_P^2 = 0.06$) and functional temporal stability ($\eta_P^2 = 0.05$). The importance of the interaction terms relative to the driver strength of management or emissions scenario alone highlights that both management and emissions scenario have context-dependent influence, and that local site characteristics modulate how management and emissions scenario impact ecosystem stability. Management alone also had a significant impact on stability, but it depended on emissions scenario, and vice versa. Finally, our analysis revealed that management and emissions scenario interacted to alter stability outcomes for temporal stability and compositional resilience.



Table 1

Strength of Management Type, Emissions Scenario (RCP), and Region Group Membership as Predictors of Ecosystem Functional Stability

			Resilience			Temporal stability		
	Predictor	Df	SS	F-value	Partial $\eta 2$	SS	F-value	Partial η^2
Function	Region	1	397.40	121.55*	0.11	35.01	2054.08*	0.68
	RCP	1	4.10	1.26	0.00	0.40	23.66*	0.02
	Management	2	36.50	5.58*	0.01	0.26	7.77*	0.02
	Region:RCP	1	24.10	7.39*	0.01	0.90	52.97*	0.05
	Region:Management	2	135.60	20.74*	0.04	0.43	12.52*	0.03
	RCP:Management	2	5.90	0.91	0.00	0.52	15.36*	0.03
	Region:RCP:Management	2	0.50	0.07	0.00	0.44	13.01*	0.03
Composition	Region	1	121.50	229.02*	0.11	119.90	318.05*	0.14
	RCP	1	16.30	30.75*	0.02	5.10	13.64*	0.01
	Management	2	15.50	14.63*	0.02	7.10	9.36*	0.01
	Region:RCP	1	67.40	127.04*	0.06	2.80	7.48*	0.00
	Region:Management	2	28.70	27.01*	0.03	16.10	21.38*	0.02
	RCP:Management	2	13.90	13.08*	0.01	0.60	0.83	0.00
	Region:RCP:Management	2	9.30	8.79*	0.01	0.00	0.02	0.00

Note. *p < 0.05. "*" Symbols next to f-values denote statistical significance at an alpha value of 0.05, and ":" symbols between predictors indicate interaction effects between categorical factors. Df = degrees of freedom and SS = sum of squares.

3.5. Managing for Enhanced Resilience

Our analysis revealed that specific structural factors affect functional and compositional resilience. We found that targeted management activities could improve resilience (Figure S4 in Supporting Information S1), but their effectiveness was regionally dependent. In the SE, trees with greater variability in height were linked to higher functional resilience, while high LAI, tree density, and crown size were associated with reduced resilience. Conversely, in the GL, high LAI, tree density, and crown size were associated with higher NPP resilience, while an abundance of large, older trees was linked to lower resilience. AGB and SOC resilience had opposing relationships in the GL and SE, suggesting trade-offs when managing for compositional resilience. In the SE, higher LAI and larger, more variable crowns were linked to enhanced AGB and SOC resilience, respectively. In the GL, managing for older, taller trees with variable DBH enhanced AGB resilience, and increased stand density, and larger crowns resulted in higher SOC resilience (Figure S4 in Supporting Information S1).

4. Discussion

This study investigated the long-term impact of forest management and climate change on the structure and function of forests in two regions, the GL and the SE U.S., over a multidecadal time frame. Forest management was represented as one of three types: even-aged, low-intensity uneven-aged, and high-intensity uneven-aged management, which were defined by regionally specific model parameters. Alternate climate futures were represented with two representative concentration pathways, RCP4.5 and RCP8.5, using downscaled CMIP5 meteorological forcing data. We assessed the impact of management and climate change on forest structure compared to a control scenario without management intervention and evaluated the response of forest functional and compositional stability, measured as resilience and temporal stability. Finally, we analyzed the drivers of the observed changes in forest stability, and outlined potential management techniques that could enhance forest resilience in the two regions.

4.1. Management and Climate Impacts on Forest Structure

The first research objective was to examine the impact of varying levels of management intensity on forest structure across multidecadal timescales and explore whether the effects were regionally dependent or climate change scenario-specific. Nine variables, including LAI, density, age and size distributions of trees, and the



structural complexity of forests, were used to represent forest structure. Tree height and size variability demonstrated strong climate sensitivity in both regions, but structural variables differed more between management types than between climate scenarios, consistent with recent work showing that anthropogenic disturbance is a stronger driver of compositional changes than climate at the century scale (Danneyrolles et al., 2019). A notable climate-driven mortality signal was observed in the GL, which saw amplified tree mortality for most management types under the higher emissions scenario.

The same management strategy often produced divergent structural outcomes over time when applied in different regions, likely due to interactions with regional environmental conditions and climate change. While management might be the primary factor shaping forest structure, climate shapes vegetation composition and functional diversity (Harrison et al., 2020; Kreft & Jetz, 2007) through temperature and moisture availability, which contribute to changes in structure as trees compete for available resources. For example, Ehbrecht et al. (2021) found that global variability in the structural complexity of undisturbed primary forests was largely determined by annual precipitation and precipitation seasonality, and Danneyrolles et al. (2019) drew connections to the implications of increased variability and magnitude of temperature and precipitation extremes under future radiative concentration pathways, showing that the probability of large compositional and structural changes in vegetation is greater under high-emission scenarios (RCP8.5). Within ED2, ecosystem structure evolves in response to tree growth, reproduction, mortality, disturbance, and aging (Longo et al., 2019), and functional diversity and its effects on competitive advantage are represented, meaning the influence of climate and interactions with management through shaping ecosystem structure can be captured.

The management-free control scenario in both regions featured large, tall, older trees with high AGB, consistent with studies on the effect of excluding management on carbon stocks (Andrews et al., 2018; Brice et al., 2020; Laflower et al., 2016). Among the management strategies, structural outcomes under even-age management differed the most, while the two uneven-age management strategies often had similar outcomes, especially in the GL. Even-age management led to dense stands of shorter and smaller trees with large crowns and low variability in diameter, which is consistent with recent studies evaluating global forest demographic responses to increased climate variability (McDowell et al., 2020; McIntyre et al., 2015). This compositional shift is likely more prominent under even-age management than under the control due to the periodic harvesting of large trees accelerating community transition. In the GL, even-age management resulted in stands of young trees with low average AGB, low structural complexity, and lower mortality. This suggests a simplification of canopy structure in response to intermittent management, which contrasts to work showing that small-scale intermittent disturbances often result in increased canopy complexity (Ehbrecht et al., 2017; Hardiman et al., 2013), for example, by creating gaps in the canopy that allow for light penetration and support understory development (Willim et al., 2019). However, this may be a consequence of the simulated disturbance having the same intensity and targeting the same PFTs and size classes for removal with each recurrence. Even-age management had similar but less pronounced effects on structural complexity in the SE, although variability in crown size increased, potentially due to the region experiencing less heat and water stress than the GL, both of which have been linked to crown dieback (Matusick et al., 2018).

Uneven-aged management is characterized by moderately tall trees with a higher LAI than the control in both regions. Structural differences between the two uneven-age management strategies were often not significant in the GL, although trees were taller and larger with lower densities under uneven-age management, and variability in height and DBH was greater. In the SE, high-intensity uneven-age management led to younger average ages, high variability in tree diameter, and low mortality, suggesting successful recruitment and survivorship. In contrast, low-intensity uneven-age management in the SE resulted in lower LAI but larger average diameters and greater structural complexity. In practice, extended rotations combined with thinning or regeneration support have been used to promote structural complexity and ecological dynamics that are more commonly observed in mature or old-growth forests (Kaarakka et al., 2021). These findings support the hypothesis that the response of forest structure to management depends on management intensity and severity and that structural responses to management will diverge regionally due to differences in regional climate change impacts.

Notably, mortality was highest under the control scenario in both regions and increased substantially by the end of the century. This suggests that management could buffer the acceleration of climate-driven mortality, although the extent to which this is observed depends on the management type and the region. Mortality in ED2 is determined by aging, carbon starvation, fire, and cold or frost events (Longo et al., 2019). Carbon starvation as a



driver of mortality is density-dependent, meaning it occurs in response to trees actively competing for shared resources such as light and water. Therefore, the decreased rates of mortality observed under active management could potentially be attributed to intermittent harvest reducing competition for limited water resources, and the substantial increases in mortality over time under the control scenario indicate that these resources increase in scarcity under climate change. For example, in the SE, uneven-aged management- which periodically removes large-diameter trees- resulted in lower mortality rates compared to even-age management and the control. However, high-intensity uneven-age management, which also involved understory thinning, had the lowest mortality rates across all emissions scenarios.

4.2. Forest Stability

The second research objective was to determine how forest management alters ecosystem stability in the face of climate change and whether relationships are regionally dependent. We hypothesized that more intensive management practices would decrease forest stability but found that stability outcomes were largely regionally dependent. For example, in the SE, high-intensity uneven-age management enhanced functional (NPP) resilience relative to the control regardless of climate change scenario, but the same strategy had the opposite effect in the GL. The higher functional resilience in the SE could be partially attributed to regional climate differences. According to the physiological tolerance hypothesis, warmer and more humid climates support a broader range of plant functional strategies, leading to greater tolerance to climatic shifts like temperature, moisture, and light (Harrison et al., 2020). Additionally, the SE experienced increased water availability under both emissions scenarios, which has been linked to accelerated recovery rates after disturbance (Anderson-Teixeira et al., 2013).

Of the three management approaches, only even-age management positively impacted functional resilience in the GL. Notably, only even-age management reduced average tree height and size relative to the control in the GL, and larger trees have a greater risk of mortality under water stress (McIntyre et al., 2015; Nepstad et al., 2007) due to hydraulic failure and carbon starvation (McDowell & Allen, 2015; Sevanto et al., 2014). For example, Stovall et al. (2019) showed that tree height was the strongest predictor of mortality during drought conditions, with large trees dying at twice the rate of small trees, and mortality risk increasing nonlinearly with environmental stress. Although hydraulic failure is not explicitly represented in ED2, both the mortality rate and the growth rate depend on tree size and carbon balance. Management impacts on compositional resilience, or the resilience of carbon stored as AGB and in soils, were also regionally dependent, with significant differences observed between management types in the SE but not in the GL. Both high-intensity uneven-age and even-age management enhanced compositional resilience relative to the control in the SE, although their effectiveness differed depending on emissions scenario. The GL experienced lower compositional resilience under the high-emissions scenario than the SE, particularly in SOC stores, suggesting greater climate sensitivity. Temperatures increased slightly in the GL, which has been linked to accelerated decomposition of organic matter and carbon compounds in the soil (Natali et al., 2014; Schuur et al., 2015; Turetsky et al., 2020). However, the GL also became more water-limited during the growing season, likely reducing SOC resilience due to decreased organic inputs in dry conditions, as evidenced by the decrease in AGB. Temporal stability analysis revealed that regional differences were most pronounced for stored carbon stocks, with higher stability observed in the GL. However, this finding can be misleading, as the magnitude of carbon stored in AGB and soils decreased over time in the GL while also becoming less variable, resulting in higher compositional temporal stability than the SE, where the magnitude of carbon stores was more variable. The high temporal variability in the SE is likely related to increased variability in environmental conditions, particularly precipitation extremes.

We hypothesized that management is the most important factor shaping functional stability but found that while management is the most important driver of functional resilience, functional temporal stability was more closely linked to emissions scenario. This is likely due to the tight coupling between primary production and environmental drivers of photosynthesis that vary latitudinally and experience significant interannual fluctuations (Desai, 2010; Desai et al., 2022). However, management and interactions between climate change and management were still significant predictors of functional temporal stability, reflecting the influence of both microand macroenvironmental conditions on NPP. For example, temperature and vapor pressure deficit can vary across large scales because of climate change, but they can also vary at small scales due to differences in the horizontal and the vertical positioning of trees, abundance of canopy gaps, etc., which are directly impacted by management. Additionally, this finding could be partially attributed to the influence of management on successional dynamics across multidecadal timescales, capturing NPP responses to shifts in community composition and competition.

Compositional resilience was primarily shaped by regional scale climate, although interactions between emissions scenarios and management were also significant drivers. This finding is consistent with the work by Felipe-Lucia et al. (2018), which found that environmental factors followed by the combination of structural attributes and environmental factors were the strongest drivers of variability in SOC in temperate forests. This suggests that while focusing mitigation efforts on enhancing functional resilience might be the most impactful, the impact of proposed mitigation strategies on compositional resilience cannot be ignored when designing mitigation plans. Importantly, the dominance of management-region interaction terms and emissions scenario-region interaction terms over the importance of management or emissions scenario alone emphasizes that both management and emissions scenario have context-dependent influence, and that local site characteristics modulate how management and emissions scenario impact ecosystem stability. The positive effects of management on functional and compositional resilience were generally more pronounced in the SE, whereas management often reduced resilience relative to the control in the GL, particularly under the high-emissions scenario. This suggests that alternate management strategies beyond those evaluated here should be explored for the GL, along with associated climate sensitivities.

Additionally, targeting specific structural attributes through management could increase functional resilience in the face of ongoing climate change, but trade-offs often exist. For example, increasing functional resilience in the SE came at the expense of SOC resilience. We also showed that management approaches to bolstering resilience must be regionally specific, as a strategy that enhances resilience in one region can hinder it in another. For example, functional resilience in the GL was enhanced by managing for larger crown areas and greater variability in crown size, potentially due to the ability of a more closed canopy to regulate forest microclimate (Felipe-Lucia et al., 2018) by reducing incoming radiation and evaporative loss. In the SE, however, these same management activities slightly negatively impacted functional resilience, and resilience was instead enhanced by managing for greater variability in tree height (Figure S4 in Supporting Information S1). The strong regional dependence of forest stability emphasizes the need to tailor management practices based on the region and consider interactions with regional climate change when designing management plans.

Finally, we hypothesized that there would be a positive correlation between forest stability and structural complexity due to demonstrated connections between structural complexity and forest productivity and biodiversity (Dănescu et al., 2016; Fahey et al., 2015; Gough et al., 2019; Hardiman et al., 2013; Ishii et al., 2004). While this was true to an extent, for example, greater variability in tree height enhanced functional resilience in the SE, enhancing structural complexity was not uniformly associated with an increase in resilience. For example, higher variability in tree diameter was correlated with decreased functional resilience in the GL. This is likely influenced by limiting our definition of structural complexity to variability in tree height, diameter, and crown size, whereas the interaction between structural attributes is an important aspect of complexity (McElhinny et al., 2005). A holistic approach to describing structural complexity, such as using a metric encapsulating the three-dimensional spatial arrangement of vegetation, as presented by Zenner and Hibbs (2000) and Ehbrecht et al. (2017), would quantify complexity more objectively and facilitate comparison between regions and management types.

4.3. Study Limitations

While we acknowledge the critical role that natural disturbances play in shaping forest dynamics (Anderegg, Chegwidden, et al., 2022; Hicke et al., 2012; Pugh et al., 2019), this study does not explicitly evaluate the impacts of natural disturbances on forest structure and stability, nor how natural disturbances might intensify with climate change and interact with management to impact forest dynamics. This decision stems from the study's primary focus on understanding how management influences forest structure and how unforeseen interactions with shifting climatic conditions might affect the ability of management to enhance forest stability. To achieve this, we intentionally isolate key variables to disentangle these relationships, as introducing too many factors simultaneously would obscure these interactions. However, with climate change increasing the frequency and intensity of natural disturbances (Seidl et al., 2017; Turner, 2010), their exclusion here limits understanding of the potential effects of enhanced climate stress on carbon dynamics. Therefore, investigating the interplay of natural disturbances with management strategies presents a valuable avenue for future research, building on the foundational insights into this study.



5. Conclusions

This study examined the carbon cycle impacts of forest management, potential interactions with climate change, and consequences for forest stability. We adopted a multidimensional approach to evaluate stability, and in contrast to many previous studies evaluating forest management as a NbCS, included forest SOC, the largest forest carbon pool, in our analysis (Kaarakka et al., 2021). For forest management to be a viable NbCS, it must account for the durability of enhanced carbon sequestration and preserve existing carbon stores (Novick et al., 2024). However, accounting for durability concerns is challenging, as the risk to existing carbon stores depends on how forests respond to climate change. Our study demonstrated that management can increase forest resilience to climate change and minimize the release of stored carbon by reducing mortality. For example, mortality rates were highest under the control scenario, while simulations under all three management scenarios resulted in reduced mortality. Our findings highlight the regional dependency of management-induced changes in forest structure and resilience, as well as the dependence of temporal stability on regional climate change. For instance, high-intensity uneven-age management, which involves short rotation harvests across a large fractional area and understory thinning, enhanced functional resilience relative to the control in the SE but had the opposite effect in the GL. This emphasizes the point that forest management NbCS approaches are not universal, and that durability must be assessed at the regional scale. This information can help forest managers evaluate trade-offs between ecosystem goods and services, assess climate risks of applying management practices in different regions, and identify specific components of ecosystem function to bolster through targeted management practices.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The meteorology downscaling code is open-source and is available through the Predictive Ecosystem Analyzer (LeBauer et al., 2013). All ED2 codes for model initialization and analysis are archived on Zenodo (Fitzpatrick & Rollinson, 2023). National Ecological Observatory Network eddy covariance flux tower instrumentation and tower information is provided in the eddy covariance bundled data product (DP4.00200.001; NEON, 2024). The AmeriFlux FLUXNET data sets for UNDE (AmeriFlux site ID US-xUN; NEON, 2023a), TALL (AmeriFlux site ID US-xTA; NEON, 2023b), and OSBS (AmeriFlux site ID US-xSB; NEON, 2023c) adhere to the Ameri-Fluxdata use policy, CC-BY-4.0 License. National Ecological Observatory Network soil and vegetation data used for model initialization include soil physical and chemical properties collected from megapits (DP1.00096.001, NEON, 2023d), soil microbial biomass (DP1.10104.001, NEON, 2023e), and vegetation structure data (DP1.10098.001, NEON, 2023f).

References

- Albani, M., Medvigy, D., Hurtt, G. C., & Moorcroft, P. R. (2006). The contributions of land-use change, CO2 fertilization, and climate variability to the Eastern US carbon sink. Global Change Biology, 12, 2370-2390. https://doi.org/10.1111/j.1365-2486.2006.01254.x
- Amiro, B. D., Barr, A. G., Barr, J. G., Black, T. A., Bracho, R., Brown, M., et al. (2010). Ecosystem carbon dioxide fluxes after disturbance in forests of North America. Journal of Geophysical Research, 115(G4). https://doi.org/10.1029/2010JG001390
- Anderegg, W. R. L., Chegwidden, O. S., Badgley, G., Trugman, A. T., Cullenward, D., Abatzoglou, J. T., et al. (2022). Future climate risks from stress, insects and fire across US forests. Ecology Letters, 25(6), 1510-1520. https://doi.org/10.1111/ele.14018
- Anderegg, W. R. L., Trugman, A. T., Wang, J., & Wu, C. (2022). Open science priorities for rigorous nature-based climate solutions. PLoS Biology, 20(12), e3001929. https://doi.org/10.1371/journal.pbio.3001929
- Anderson-Teixeira, K. J., Miller, A. D., Mohan, J. E., Hudiburg, T. W., Duval, B. D., & DeLucia, E. H. (2013). Altered dynamics of forest recovery under a changing climate. Global Change Biology, 19(7), 2001-2021. https://doi.org/10.1111/gcb.12194
- Andrews, T., Dietze, M., & Booth, R. (2018). Climate or disturbance: Temperate forest structural change and carbon sink potential. bioRxiv, 478693. https://doi.org/10.1101/478693
- Atkins, J. W., Bond-Lamberty, B., Fahey, R. T., Haber, L. T., Stuart-Haëntjens, E., Hardiman, B. S., et al. (2020). Application of multidimensional structural characterization to detect and describe moderate forest disturbance. Ecosphere, 11(6), e03156. https://doi.org/10.1002/ecs2.3156
- Atkins, J. W., Fahey, R. T., Hardiman, B. H., & Gough, C. M. (2018). Forest canopy structural complexity and light absorption relationships at the subcontinental scale. Journal of Geophysical Research: Biogeosciences, 123(4), 1387–1405. https://doi.org/10.1002/2017JG004256
- Bacastow, R., & Keeling, C. K. (1973). Atmospheric carbon dioxide and radiocarbon in the natural carbon cycle: II. Changes from A. D. 1700 to 2070 as deduced from a geochemical model. Brookhaven Symposia in Biology, 30, 86-135.
- Bauman, D., Fortunel, C., Delhaye, G., Malhi, Y., Cernusak, L. A., Bentley, L. P., et al. (2022). Tropical tree mortality has increased with rising atmospheric water stress. Nature, 608(7923), 528-533. https://doi.org/10.1038/s41586-022-04737-7
- Becknell, J. M., Desai, A. R., Dietze, M. C., Schultz, C. A., Starr, G., Duffy, P. A., et al. (2015). Assessing interactions among changing climate, management, and disturbance in forests: A Macrosystems approach. BioScience, 65(3), 263-274. https://doi.org/10.1093/biosci/biu234

This work was primarily supported by the NSF Emerging Frontiers Macrosystems Biology program award DEB-1702996 through the Management and Disturbance in Forest Ecosystems (MANDIFORE) project (Co-PIs Ankur Desai, Christina Staudhammer, Michael Dietze, Paul Duffy, Courtney Schultz, William Kleindl, and Michael Binford). Additional funding support for Bailey Murphy was provided through the Next Generation Ecosystem Experiment Arctic (NGEE Arctic) project. The NGEE Arctic project is supported by the Office of Biological and Environmental Research in the U.S. Department of Energy's Office of Science. The authors would like to thank Eliisa Carter for her comments that helped improve the manuscript, and Paul Duffy of Neptune and Company for synthesizing management information from interviews with forestry experts into ED2 model parameter values. This manuscript has been authored in part by UT-Battelle, LLC, under contract DE-AC05-00OR22725 with the U.S. Department of Energy (DOE). The U.S. government retains and the publisher, by accepting the article for publication, acknowledges that the U.S. government retains a nonexclusive, paid up, irrevocable, worldwide license to publish or reproduce the published form of this manuscript, or allow others to do so, for U.S. government purposes. DOE will provide public access to these results of federally sponsored research in accordance with the public access plan (https://www. energy.gov/doe-public-access-plan).

-and-conditions) on Wiley Online Library for rules of use; OA articles

are governed by the applicable Creative Commons License

21698961, 2025, 3, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024/G008391, Wiley Online Library on [03/06/2025]. See the Terms and Conditions (https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024/G008391, Wiley Online Library on [03/06/2025]. See the Terms and Conditions (https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024/G008391, Wiley Online Library on [03/06/2025]. See the Terms and Conditions (https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2024/G008391, Wiley Online Library on [03/06/2025].



- Brice, M.-H., Vissault, S., Vieira, W., Gravel, D., Legendre, P., & Fortin, M.-J. (2020). Moderate disturbances accelerate forest transition dynamics under climate change in the temperate–boreal ecotone of eastern North America. *Global Change Biology*, 26(8), 4418–4435. https://doi. org/10.1111/gcb.15143
- Canadell, J. G., & Raupach, M. R. (2008). Managing forests for climate change mitigation. *Science*, 320(5882), 1456–1457. https://doi.org/10. 1126/science.1155458
- Canadell, J. G., & Schulze, E. D. (2014). Global potential of biospheric carbon management for climate mitigation. Nature Communications, 5(1), 5282. Article 1. https://doi.org/10.1038/ncomms6282
- Carey, A. B. (2006). Active and passive forest management for multiple values. Northwestern Naturalist, 87(1), 18. https://doi.org/10.1898/1051-1733(2006)87[18:AAPFMF]2.0.CO;2
- Ciais, P., Sabine, C., Bala, G., Bopp, L., Brovkin, V., Canadell, J., et al. (2013). Carbon and other biogeochemical cycles. In *Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change.* Cambridge University Press.
- Dănescu, A., Albrecht, A. T., & Bauhus, J. (2016). Structural diversity promotes productivity of mixed, uneven-aged forests in southwestern Germany. Oecologia, 182(2), 319–333. https://doi.org/10.1007/s00442-016-3623-4
- Danneyrolles, V., Dupuis, S., Fortin, G., Leroyer, M., de Römer, A., Terrail, R., et al. (2019). Stronger influence of anthropogenic disturbance than climate change on century-scale compositional changes in northern forests. *Nature Communications*, 10(1), 1265. Article 1. https://doi.org/10. 1038/s41467-019-09265-z
- De Kauwe, M. G., Medlyn, B. E., Zaehle, S., Walker, A. P., Dietze, M. C., Hickler, T., et al. (2013). Forest water use and water use efficiency at elevated CO2: A model-data intercomparison at two contrasting temperate forest face sites. *Global Change Biology*, 19(6), 1759–1779. https:// doi.org/10.1111/gcb.12164
- De Kauwe, M. G., Medlyn, B. E., Zaehle, S., Walker, A. P., Dietze, M. C., Wang, Y.-P., et al. (2014). Where does the carbon go? A model–data intercomparison of vegetation carbon allocation and turnover processes at two temperate forest free-air CO2 enrichment sites. *New Phytologist*, 203(3), 883–899. https://doi.org/10.1111/nph.12847
- Desai, A. R. (2010). Climatic and phenological controls on coherent regional interannual variability of carbon dioxide flux in a heterogeneous landscape. *Journal of Geophysical Research*, 115(G3), G00J02. https://doi.org/10.1029/2010JG001423
- Desai, A. R., Murphy, B. A., Wiesner, S., Thom, J., Butterworth, B. J., Koupaei-Abyazani, N., et al. (2022). Drivers of decadal carbon fluxes across temperate ecosystems. *Journal of Geophysical Research: Biogeosciences*, 127(12), e2022JG007014. https://doi.org/10.1029/ 2022JG007014
- Dixon, G. E. (2002). Essential FVS: A user's guide to the forest vegetation simulator. Internal rep (Vol. 226p). U. S. Department of Agriculture, Forest Service, Forest Management Service Center. (Revised: August 10, 2023).
- Dorheim, K., Gough, C. M., Haber, L. T., Mathes, K. C., Shiklomanov, A. N., & Bond-Lamberty, B. (2022). Climate drives modeled forest carbon cycling resistance and resilience in the upper Great lakes region, USA. *Journal of Geophysical Research: Biogeosciences*, 127(1), e2021JG006587. https://doi.org/10.1029/2021JG006587
- Ehbrecht, M., Schall, P., Ammer, C., & Seidel, D. (2017). Quantifying stand structural complexity and its relationship with forest management, tree species diversity and microclimate. Agricultural and Forest Meteorology, 242, 1–9. https://doi.org/10.1016/j.agrformet.2017.04.012
- Ehbrecht, M., Seidel, D., Annighöfer, P., Kreft, H., Köhler, M., Zemp, D. C., et al. (2021). Global patterns and climatic controls of forest structural complexity. *Nature Communications*, 12(1), 519. https://doi.org/10.1038/s41467-020-20767-z
- Fahey, R. T., Fotis, A. T., & Woods, K. D. (2015). Quantifying canopy complexity and effects on productivity and resilience in late-successional hemlock-hardwood forests. *Ecological Applications : A Publication of the Ecological Society of America*, 25(3), 834–847. https://doi.org/10. 1890/14-1012.1
- Fargione, J. E., Bassett, S., Boucher, T., Bridgham, S. D., Conant, R. T., Cook-Patton, S. C., et al. (2018). Natural climate solutions for the United States. Science Advances, 4(11), eaat1869. https://doi.org/10.1126/sciadv.aat1869
- Felipe-Lucia, M. R., Soliveres, S., Penone, C., Manning, P., van der Plas, F., Boch, S., et al. (2018). Multiple forest attributes underpin the supply of multiple ecosystem services. *Nature Communications*, 9(1), 4839. https://doi.org/10.1038/s41467-018-07082-4
- Fisher, R. A., & Koven, C. D. (2020). Perspectives on the future of land surface models and the challenges of representing complex terrestrial systems. *Journal of Advances in Modeling Earth Systems*, 12(4), e2018MS001453. https://doi.org/10.1029/2018MS001453
- Fitzpatrick, L., & Rollinson, C. (2023). MortonArb-ForestEcology/MortonArb- ManagementxClimate-Modeling: Code associated with first submission of peer-reviewed manuscript (Version Manuscript-v1). Zenodo. [Computer software]. https://doi.org/10.5281/zenodo.8211973
- Forzieri, G., Dakos, V., McDowell, N. G., Ramdane, A., & Cescatti, A. (2022). Emerging signals of declining forest resilience under climate change. *Nature*, 608(7923), 534–539. https://doi.org/10.1038/s41586-022-04959-9
- Fuss, S., Canadell, J. G., Peters, G. P., Tavoni, M., Andrew, R. M., Ciais, P., et al. (2014). Betting on negative emissions. *Nature Climate Change*, 4(10), 850–853. Article 10. https://doi.org/10.1038/nclimate2392
- Gough, C. M., Atkins, J. W., Fahey, R. T., & Hardiman, B. S. (2019). High rates of primary production in structurally complex forests. *Ecology*, 100(10), e02864. https://doi.org/10.1002/ecy.2864
- Gough, C. M., Hardiman, B. S., Nave, L. E., Bohrer, G., Maurer, K. D., Vogel, C. S., et al. (2013). Sustained carbon uptake and storage following moderate disturbance in a Great Lakes forest. *Ecological Applications : A Publication of the Ecological Society of America*, 23(5), 1202–1215. https://doi.org/10.1890/12-1554.1
- Hardiman, B. S., Bohrer, G., Gough, C. M., & Curtis, P. S. (2013). Canopy structural changes following widespread mortality of canopy dominant trees. *Forests*, 4(3), 537–552. Article 3. https://doi.org/10.3390/f4030537
- Harrison, S., Spasojevic, M. J., & Li, D. (2020). Climate and plant community diversity in space and time. Proceedings of the National Academy of Sciences, 117(9), 4464–4470. https://doi.org/10.1073/pnas.1921724117
- Hatcher, C. (2017). NEON site level plot summary Talladega National Forest. National Ecological Observatory Network.
- Hicke, J. A., Allen, C. D., Desai, A. R., Dietze, M. C., Hall, R. J., Ted Hogg, E. H., et al. (2012). Effects of biotic disturbances on forest carbon cycling in the United States and Canada. *Global Change Biology*, *18*(1), 7–34. https://doi.org/10.1111/j.1365-2486.2011.02543.x
- Hillebrand, H., Langenheder, S., Lebret, K., Lindström, E., Östman, Ö., & Striebel, M. (2018). Decomposing multiple dimensions of stability in global change experiments. *Ecology Letters*, 21(1), 21–30. https://doi.org/10.1111/ele.12867
- Holling, C. S. (1973). Resilience and stability of ecological systems. Annual Review of Ecology and Systematics, 4(1), 1–23. https://doi.org/10. 1146/annurev.es.04.110173.000245
- Hurteau, M. D., Bradford, J. B., Fulé, P. Z., Taylor, A. H., & Martin, K. L. (2014). Climate change, fire management, and ecological services in the southwestern US. Forest Ecology and Management, 327, 280–289. https://doi.org/10.1016/j.foreco.2013.08.007

- Hurtt, G. C., Pacala, S. W., Moorcroft, P. R., Caspersen, J., Shevliakova, E., Houghton, R. A., & Moore, B. (2002). Projecting the future of the U.S. carbon sink. *Proceedings of the National Academy of Sciences*, 99(3), 1389–1394. https://doi.org/10.1073/pnas.012249999
- IPCC. (2022). Climate change 2022: Mitigation of climate change. Contribution of working group III to the sixth assessment report of the intergovernmental panel on climate change. Cambridge University Press.
- Ishii, H. T., Tanabe, S., & Hiura, T. (2004). Exploring the relationships among canopy structure, stand productivity, and biodiversity of temperate forest ecosystems. *Forest Science*, 50(3), 342–355. https://doi.org/10.1093/forestscience/50.3.342
- Kaarakka, L., Cornett, M., Domke, G., Ontl, T., & Dee, L. E. (2021). Improved forest management as a natural climate solution: A review. *Ecological Solutions and Evidence*, 2(3), e12090. https://doi.org/10.1002/2688-8319.12090
- Krauss, R. (2018). Terrestrial Observation System (TOS) site characterization report: Domain 05. NEON. (DOC.003889vB).
- Kreft, H., & Jetz, W. (2007). Global patterns and determinants of vascular plant diversity. Proceedings of the National Academy of Sciences of the United States of America, 104(14), 5925–5930. https://doi.org/10.1073/pnas.0608361104
- Laflower, D. M., Hurteau, M. D., Koch, G. W., North, M. P., & Hungate, B. A. (2016). Climate-driven changes in forest succession and the influence of management on forest carbon dynamics in the Puget Lowlands of Washington State, USA. *Forest Ecology and Management*, 362, 194–204. https://doi.org/10.1016/j.foreco.2015.12.015
- LeBauer, D., Wang, D., Ritcher, K. T., Davidson, C. D., & Dietze, M. C. (2013). PEcAn: The predictive ecosystem analyzer Version 1.1 [Computer software].https://www.ideals.illinois.edu/items/34890
- Lehman, C. L., & Tilman, D. (2000). Biodiversity, stability, and productivity in competitive communities. *The American Naturalist*, 156(5), 534– 552. https://doi.org/10.1086/303402
- Livingston, S. (2014). 9,700 acres tell stories of Florida's past. Florida Museum News and Blogs. https://www.floridamuseum.ufl.edu/science/ 9700-acres-tell-stories-of-floridas-past/
- Longo, M., Knox, R. G., Medvigy, D. M., Levine, N. M., Dietze, M. C., Kim, Y., et al. (2019). The biophysics, ecology, and biogeochemistry of functionally diverse, vertically and horizontally heterogeneous ecosystems: The Ecosystem Demography model, version 2.2 – Part 1: Model description. *Geoscientific Model Development*, 12(10), 4309–4346. https://doi.org/10.5194/gmd-12-4309-2019
- Mahon, B. (2003). A clearcutting history survey of the UNDERC property. University of Notre Dame.
- Matheny, A. M., Bohrer, G., Vogel, C. S., Morin, T. H., He, L., Frasson, R. P. d. M., et al. (2014). Species-specific transpiration responses to intermediate disturbance in a northern hardwood forest. *Journal of Geophysical Research: Biogeosciences*, 119(12), 2292–2311. https://doi. org/10.1002/2014JG002804
- Mathes, K. C., Ju, Y., Kleinke, C., Oldfield, C., Bohrer, G., Bond-Lamberty, B., et al. (2021). A multidimensional stability framework enhances interpretation and comparison of carbon cycling response to disturbance. *Ecosphere*, 12(11), e03800. https://doi.org/10.1002/ecs2.3800
- Matusick, G., Ruthrof, K. X., Kala, J., Brouwers, N. C., Breshears, D. D., & Hardy, G. E. S. J. (2018). Chronic historical drought legacy exacerbates tree mortality and crown dieback during acute heatwave-compounded drought. *Environmental Research Letters*, 13(9), 095002. https://doi.org/10.1088/1748-9326/aad8cb
- McDowell, N. G., & Allen, C. D. (2015). Darcy's law predicts widespread forest mortality under climate warming. *Nature Climate Change*, 5(7), 669–672. https://doi.org/10.1038/nclimate2641
- McDowell, N. G., Allen, C. D., Anderson-Teixeira, K., Aukema, B. H., Bond-Lamberty, B., Chini, L., et al. (2020). Pervasive shifts in forest dynamics in a changing world. *Science*, 368(6494), eaaz9463. https://doi.org/10.1126/science.aaz9463
- McElhinny, C., Gibbons, P., Brack, C., & Bauhus, J. (2005). Forest and woodland stand structural complexity: Its definition and measurement. *Forest Ecology and Management*, 218(1–3), 1–24. https://doi.org/10.1016/j.foreco.2005.08.034
- McIntyre, P. J., Thorne, J. H., Dolanc, C. R., Flint, A. L., Flint, L. E., Kelly, M., & Ackerly, D. D. (2015). Twentieth-century shifts in forest structure in California: Denser forests, smaller trees, and increased dominance of oaks. *Proceedings of the National Academy of Sciences*, 112(5), 1458–1463. https://doi.org/10.1073/pnas.1410186112
- Medvigy, D., Wofsy, S. C., Munger, J. W., Hollinger, D. Y., & Moorcroft, P. R. (2009). Mechanistic scaling of ecosystem function and dynamics in space and time: Ecosystem Demography model version 2. *Journal of Geophysical Research*, 114(G1). https://doi.org/10.1029/ 2008JG000812
- Moorcroft, P. R., Hurtt, G. C., & Pacala, S. W. (2001). A method for scaling vegetation dynamics: The ecosystem demography model (ed). *Ecological Monographs*, 71(4), 557–586. https://doi.org/10.1890/0012-9615(2001)071[0557:AMFSVD]2.0.CO;2
- Murphy, B. A., May, J. A., Butterworth, B. J., Andresen, C. G., & Desai, A. R. (2022). Unraveling forest complexity: Resource use efficiency, disturbance, and the structure-function relationship. *Journal of Geophysical Research: Biogeosciences*, 127(6), e2021JG006748. https://doi. org/10.1029/2021JG006748
- Natali, S. M., Schuur, E. A. G., Webb, E. E., Pries, C. E. H., & Crummer, K. G. (2014). Permafrost degradation stimulates carbon loss from experimentally warmed tundra. *Ecology*, 95(3), 602–608. https://doi.org/10.1890/13-0602.1
- National Academies of Sciences, Engineering, and Medicine. (2019). Negative emissions technologies and reliable sequestration: A research agenda. The National Academies Press. https://doi.org/10.17226/25259
- Nelson, J. A., Carvalhais, N., Migliavacca, M., Reichstein, M., & Jung, M. (2018). Water-stress-induced breakdown of carbon-water relations: Indicators from diurnal FLUXNET patterns. *Biogeosciences*, 15(8), 2433–2447. https://doi.org/10.5194/bg-15-2433-2018
- NEON (National Ecological Observatory Network). (2023c). AmeriFlux FLUXNET-1F US-xSB NEON ordway-swisher biological Station (OSBS) Ver. 3-5. AmeriFlux AMP. [Dataset]. https://doi.org/10.17190/AMF/1985451
- NEON (National Ecological Observatory Network). (2023b). AmeriFlux FLUXNET-1F US-xTA NEON Talladega national forest (TALL) Ver. 3-5. AmeriFlux AMP. [Dataset]. https://doi.org/10.17190/AMF/1985455
- NEON (National Ecological Observatory Network). (2023a). AmeriFlux FLUXNET-1F US-xUN NEON University of Notre Dame Environmental Research Center (UNDE) Ver. 3-5. AmeriFlux AMP. [Dataset]. https://doi.org/10.17190/AMF/1985458
- NEON (National Ecological Observatory Network). (2023e). Soil microbe biomass (DP1.10104.001). National Ecological Observatory Network (NEON). [Dataset]. https://doi.org/10.48443/RWBJ-RY66
- NEON (National Ecological Observatory Network). (2023d). Soil physical and chemical properties, Megapit (DP1.00096.001). National Ecological Observatory Network (NEON). [Dataset]. https://doi.org/10.48443/T70Z-NP08
- NEON (National Ecological Observatory Network). (2023f). Vegetation structure (DP1.10098.001). National Ecological Observatory Network (NEON). [Dataset]. https://doi.org/10.48443/73ZN-K414
- NEON (National Ecological Observatory Network). (2024). Bundled data products Eddy covariance (DP4.00200.001), [Dataset]RELEASE-2024 Retrieved from https://doi.org/10.48443/j9pt-m241. accessed from February 27, 2024.



- Nepstad, D. C., Tohver, I. M., Ray, D., Moutinho, P., & Cardinot, G. (2007). Mortality of large trees and lianas following experimental drought in an Amazon forest. *Ecology*, 88(9), 2259–2269. https://doi.org/10.1890/06-1046.1
- Novick, K., Williams, C., Rankle, B., Anderegg, W., Hollinger, D., Litvak, M., et al. (2022). The science needed for robust, scalable, and credible nature-based climate solutions in the United States: Summary Report. *IUScholarWorks*. https://doi.org/10.5967/8RGP-TC11
- Novick, K. A., Keenan, T. F., Anderegg, W. R. L., Normile, C. P., Runkle, B. R. K., Oldfield, E. E., et al. (2024). We need a solid scientific basis for nature-based climate solutions in the United States. *Proceedings of the National Academy of Sciences*, 121(14), e2318505121. https://doi. org/10.1073/pnas.2318505121
- Parsley, J. (2016). NEON site level plot summary university of Notre Dame environmental research center. National Ecological Observatory Network.
- Pasquill, R. G. (2006). Historic notes on the Oakmulgee division of the Talladega national forest. https://www.fs.usda.gov/Internet/FSE_ DOCUMENTS/fseprd587006.pdf
- Prink, C., & Figueroa, M. (2019). NEON site level plot summary Ordway-Swisher Biological Station. *National Ecological Observatory Network*. Pugh, T. A. M., Arneth, A., Kautz, M., Poulter, B., & Smith, B. (2019). Important role of forest disturbances in the global biomass turnover and carbon sinks. *Nature Geoscience*, 12(9), 730–735. https://doi.org/10.1038/s41561-019-0427-2
- R Core Team. (2021). R: A language and environment for statistical computing (4.0.4). R Foundation for Statistical Computing. [Computer software] Retrieved from https://www.R-project.org/
- Reich, P. B., Bermudez, R., Montgomery, R. A., Rich, R. L., Rice, K. E., Hobbie, S. E., & Stefanski, A. (2022). Even modest climate change may lead to major transitions in boreal forests. *Nature*, 608(7923), 540–545. https://doi.org/10.1038/s41586-022-05076-3
- Reichstein, M., Bahn, M., Ciais, P., Frank, D., Mahecha, M. D., Seneviratne, S. I., et al. (2013). Climate extremes and the carbon cycle. *Nature*, 500(7462), 287–295. Article 7462. https://doi.org/10.1038/nature12350
- Rockström, J., Beringer, T., Hole, D., Griscom, B., Mascia, M. B., Folke, C., & Creutzig, F. (2021). We need biosphere stewardship that protects carbon sinks and builds resilience. *Proceedings of the National Academy of Sciences*, 118(38), e2115218118. https://doi.org/10.1073/pnas. 2115218118
- Rollinson, C. R., Liu, Y., Raiho, A., Moore, D. J. P., McLachlan, J., Bishop, D. A., et al. (2017). Emergent climate and CO2 sensitivities of net primary productivity in ecosystem models do not agree with empirical data in temperate forests of eastern North America. *Global Change Biology*, 23(7), 2755–2767. https://doi.org/10.1111/gcb.13626
- Running, S. W. (2008). Ecosystem disturbance, carbon, and climate. *Science*, 321(5889), 652–653. https://doi.org/10.1126/science.1159607 Schimel, D., Hargrove, W., Hoffman, F., & MacMahon, J. (2007). NEON: A hierarchically designed national ecological network. *Frontiers in*
- *Ecology and the Environment*, 5(2), 59. https://doi.org/10.1890/1540-9295(2007)5[59:NAHDNE]2.0.CO;2
- Schuur, E. A. G., McGuire, A. D., Schädel, C., Grosse, G., Harden, J. W., Hayes, D. J., et al. (2015). Climate change and the permafrost carbon feedback. *Nature*, 520(7546), 171–179. https://doi.org/10.1038/nature14338
- Seidl, R., Thom, D., Kautz, M., Martin-Benito, D., Peltoniemi, M., Vacchiano, G., et al. (2017). Forest disturbances under climate change. Nature Climate Change, 7(6), 395–402. Article 6. https://doi.org/10.1038/nclimate3303
- Sevanto, S., McDowell, N. G., Dickman, L. T., Pangle, R., & Pockman, W. T. (2014). How do trees die? A test of the hydraulic failure and carbon starvation hypotheses. *Plant, Cell and Environment*, 37(1), 153–161. https://doi.org/10.1111/pce.12141

Simkins, J. (2017). Improving carbon cycle uncertainty through ensemble based temporal downscaling. University of Wisconsin-Madison. Smith, P., Davis, S. J., Creutzig, F., Fuss, S., Minx, J., Gabrielle, B., et al. (2016). Biophysical and economic limits to negative CO2 emissions. Nature Climate Change. 6(1), 42–50. Article 1, https://doi.org/10.1038/nclimate2870

- Stovall, A. E. L., Shugari, H., & Yang, X. (2019). Tree height explains mortality risk during an intense drought. *Nature Communications*, 10(1), 4385. https://doi.org/10.1038/s41467-019-12380-6
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Tilman, D. (1999). The ecological consequences of changes in biodiversity: A search for general principles. *Ecology*, 80(5), 1455–1474. https://doi.org/10.1890/0012-9658(1999)080[1455:TECOCI]2.0.CO;2
- Tilman, D., Reich, P. B., & Knops, J. M. H. (2006). Biodiversity and ecosystem stability in a decade-long grassland experiment. Nature, 441(7093), 629–632. https://doi.org/10.1038/nature04742
- Turetsky, M. R., Abbott, B. W., Jones, M. C., Anthony, K. W., Olefeldt, D., Schuur, E. A. G., et al. (2020). Carbon release through abrupt permafrost thaw. *Nature Geoscience*, 13(2), 138–143. https://doi.org/10.1038/s41561-019-0526-0
- Turner, M. G. (2010). Disturbance and landscape dynamics in a changing world. *Ecology*, 91(10), 2833–2849. https://doi.org/10.1890/10-0097.1 USGCRP. (2018). Impacts, risks, and adaptation in the United States: Fourth national climate assessment. *United States Global Change Research*
- Program, 2. https://doi.org/10.7930/NCA4.2018 van Vuuren, D. P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., et al. (2011). The representative concentration pathways: An
- overview. Climatic Change, 109(1–2), 5–31. https://doi.org/10.1007/s10584-011-0148-z
 von Hedemann, N., & Schultz, C. A. (2021). U.S. Family forest owners' forest management for climate adaptation: Perspectives from extension and outreach specialists. Frontiers in Climate, 3. https://doi.org/10.3389/fclim.2021.674718
- Walker, A. P., De Kauwe, M. G., Bastos, A., Belmecheri, S., Georgiou, K., Keeling, R. F., et al. (2020). Integrating the evidence for a terrestrial carbon sink caused by increasing atmospheric CO2. New Phytologist, 229(5), 2413–2445. https://doi.org/10.1111/nph.16866
- Willim, K., Stiers, M., Annighöfer, P., Ammer, C., Ehbrecht, M., Kabal, M., et al. (2019). Assessing understory complexity in beech-dominated forests (Fagus sylvatica L.) in central Europe—From managed to primary forests. *Sensors*, 19(7), 1684. https://doi.org/10.3390/s19071684
- Wood, S. N. (2011). Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models. Journal of the Royal Statistical Society - Series B: Statistical Methodology, 73(1), 3–36. https://doi.org/10.1111/j.1467-9868.2010.00749.x
- Wu, C., Coffield, S. R., Goulden, M. L., Randerson, J. T., Trugman, A. T., & Anderegg, W. R. L. (2023). Uncertainty in US forest carbon storage potential due to climate risks. *Nature Geoscience*, 16(5), 422–429. https://doi.org/10.1038/s41561-023-01166-7
- Wullschleger, S. D., Epstein, H. E., Box, E. O., Euskirchen, E. S., Goswami, S., Iversen, C. M., et al. (2014). Plant functional types in Earth system models: Past experiences and future directions for application of dynamic vegetation models in high-latitude ecosystems. *Annals of Botany*, 114(1), 1–16. https://doi.org/10.1093/aob/mcu077
- Zaehle, S., Medlyn, B. E., De Kauwe, M. G., Walker, A. P., Dietze, M. C., Hickler, T., et al. (2014). Evaluation of 11 terrestrial carbon–nitrogen cycle models against observations from two temperate Free-Air CO2 Enrichment studies. *New Phytologist*, 202(3), 803–822. https://doi.org/10. 1111/nph.12697



Zenner, E. K., & Hibbs, D. E. (2000). A new method for modeling the heterogeneity of forest structure. *Forest Ecology and Management*, 129(1), 75–87. https://doi.org/10.1016/S0378-1127(99)00140-1

References From the Supporting Information

- Pastorello, G., Trotta, C., Canfora, E., Chu, H., Christianson, D., Cheah, Y.-W., et al. (2020). The FLUXNET2015 dataset and the ONEFlux processing pipeline for eddy covariance data. *Scientific Data*, 7(1), 225. Article 1. https://doi.org/10.1038/s41597-020-0534-3
- Reichstein, M., Falge, E., Baldocchi, D., Papale, D., Aubinet, M., Berbigier, P., et al. (2005). On the separation of net ecosystem exchange into assimilation and ecosystem respiration: Review and improved algorithm. *Global Change Biology*, 11(9), 1424–1439. https://doi.org/10.1111/j. 1365-2486.2005.001002.x